Author's response to reviews

Title: The search for valved conduit tissue grafts for adults (>22mm): An ultrasonographic study of jugular vein diameters of horses and cattle

Authors:

Colin C Schwarzwald (cschwarzwald@vetclinics.uzh.ch)
Rolf Jenni (rolf_jenni@bluewin.ch)

Version: 4 Date: 12 August 2009

Author's response to reviews: see over
August 12, 2009

Re: The search for valved conduit tissue grafts for adults (>22mm): An ultrasonographic study of jugular vein diameters of horses and cattle. Schwarzwald CC, Jenni R.

Dear Editor,

We are delighted that our manuscript has been accepted for publication in your journal. The required formatting changes have been included in this final version of the manuscript.

Please do not hesitate to contact us if you have additional questions or concerns.

Sincerely,

Colin C. Schwarzwald
Dr. med. vet., PhD, Dipl. ACVIM